Author's response to reviews

Title: African-American inflammatory bowel disease in a Southern U.S. health center

Authors:

Hemanth Veluswamy (h.veluswamy@yahoo.com)
Kunal M Suryawala (ksuryawala@gmail.com)
Ankur Sheth (esalva@lsuhsc.edu)
Shannon R Wells (asheth@lsuhsc.edu)
Eric Salvatierra (swell1@lsuhsc.edu)
Walter Cromer (wcrome@lsuhsc.edu)
Ganta V Chaitanya (vganta@lsuhsc.edu)
Annette Painter (apaint@lsuhsc.edu)
Mihir Patel (patel_mihir36@yahoo.com)
Kenneth Manas (kmanas@lsuhsc.edu)
Ellen Zwank (ezwank@lsuhsc.edu)
Moheb Boktor (mbokto@lsuhsc.edu)
Konal M Baig (kbaig@lsuhsc.edu)
Balaji Datti (bdatti@lsuhsc.edu)
J Michael Mathis (jmathi@lsuhsc.edu)
Alireza Minagar (aminag@lsuhsc.edu)
Paul A Jordan (pjorda1@lsuhsc.edu)
Jonathan S Alexander (jalexa@lsuhsc.edu)

Version: 11 Date: 8 September 2010

Author's response to reviews:

typo